Author's response to reviews

Title: Tissue detection of natural killer(NK) cells in colorectal adenocarcinoma.

Authors:

Ioannis S Papanikolaou (ispapn@otenet.gr)
Andreas CH. Lazaris (alazaris@med.uoa.gr)
Periklis Apostolopoulos (periklisapo@yahoo.com)
Nikos Kavantzazs (panatomy@med.uoa.gr)
Maria G Papas (drmariapapas@hotmail.com)
Christos Mavrogiannis (chrismav@otenet.gr)
E S Patsouris (panatomy@med.uoa.gr)
Athanasios Archimandritis (aarchima@med.uoa.gr)

Version: 2 Date: 27 July 2004

Author's response to reviews:

We are honestly grateful for your reviewer suggestions concerning our paper untitled: "TISSUE DETECTION OF NATURAL KILLER CELLS IN COLORECTAL ADENOCARCINOMA" Please find below a point by point response to the reviewer comments.

In the meantime we did our best to comply accordingly.

Reply to Dr. Jesse Lachter

1. We made extensive changes in the English language throughout the text.

2. We tried to present our data in a much shortened form (mainly by shortening the Discussion and the changes are underlined into the text).

Thanking you in advance.

Sincerely

Dr. Papanikolaou - Dr. Apostolopoulos